Author's response to reviews

Title: Ethnobotanical and antimicrobial study of some selected medicinal plants used in Khyber Pakhtunkhwa (KPK) as a potential source to cure infectious diseases

Authors:

Nadeem Khan (khan_m_nadeem@yahoo.com)
Arshad M Abbasi (amabbasi@ciit.net.pk)
Ghulam Dastagir (dastagirbotany@yahoo.com)
Munir H Shah (mhshahg@hotmail.com)

Version: 2
Date: 13 December 2013

Author's response to reviews:

The voucher specimen for each plant was submitted in the herbarium of Quaid-i-Azam University Islamabad, Pakistan.
The voucher number for each specimen is listed in the table 1.